Author's response to reviews

Title: Prevalence of Prehypertension and its Relationship to Risk Factors for Cardiovascular Disease in Jamaica: Analysis from a Cross-Sectional Survey

Authors:

Trevor S Ferguson (trevor.ferguson02@uwimona.edu.jm)  
Novie O Younger (novie.younger@uwimona.edu.jm)  
Marshall K Tulloch-Reid (marshall.tullochreid@uwimona.edu.jm)  
Marilyn B Lawrence Wright (marilyn.lawrencewright@uwimona.edu.jm)  
Elizabeth M Ward (warde@moh.gov.jm)  
Deanna E Ashley (deca@cwjamaica.com)  
Rainford J Wilks (rainfordw@cwjamaica.com)

Version: 4 Date: 30 July 2008

Author's response to reviews:

July 30, 2008

Dr. Melissa Norton, MD  
Editor in Chief, BMC Cardiovascular Disorders  
BioMed Central Ltd  
Science Navigation Group  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB, United Kingdom

Dear Dr. Norton

Re: Manuscript #1589799913176674  
Prevalence of Prehypertension and its Relationship to Risk Factors for Cardiovascular Disease in Jamaica: analysis from a Cross-Sectional Survey

Attached please find our revised manuscript.

As requested, we have included sections on acknowledgements, author support and role of funding agencies.

We thank the Journal for accepting our manuscript and look forward to its publication.

Yours truly,

Rainford Wilks.